

Preprints are preliminary reports that have not undergone peer review. They should not be considered conclusive, used to inform clinical practice, or referenced by the media as validated information.

Patient and public involvement prior to trial initiation: lessons learnt for rapid partnership in the COVID-19 era.

Zahra Jamal (Zahra.jamal@lshtm.ac.uk)

London School of Hygiene & Tropical Medicine

Alexander Perkins

London School of Hygiene & Tropical Medicine

Christopher Allen

St Thomas' Hospital

Research Advisory Group

London School of Hygiene & Tropical Medicine

Richard Evans

London School of Hygiene & Tropical Medicine

Joanna Sturgess

London School of Hygiene & Tropical Medicine

Claire Snowdon

London School of Hygiene & Tropical Medicine

Tim Clayton

London School of Hygiene & Tropical Medicine

Diana Elbourne

London School of Hygiene & Tropical Medicine

Research Article

Keywords: Patient and public involvement, Clinical trials, COVID-19

Posted Date: November 11th, 2020

DOI: https://doi.org/10.21203/rs.3.rs-102043/v1

License: (a) This work is licensed under a Creative Commons Attribution 4.0 International License. Read Full License

Version of Record: A version of this preprint was published at Research Involvement and Engagement on March 8th, 2021. See the published version at https://doi.org/10.1186/s40900-021-00250-9.

Abstract

Background: Clinical trials are playing a critical role in the global public health response to the COVID-19 pandemic. Despite the increasing recognition of the value of PPI in clinical trials, just 20% of the COVID-19 research proposals reviewed by Research Ethics Committees in the UK at the start of the pandemic reported PPI. There is a perception that PPI might result in delays in delivering research and therefore delays in obtaining important results. In this paper, we report our experience of rapid PPI for a COVID-19 clinical trial.

Methods: RAPID-19 is a COVID-19 clinical trial which was planned to be submitted for fast-track ethics review in the United Kingdom. During the development of the trial protocol, the PPI Panel at the London School of Hygiene & Tropical Medicine Clinical Trials Unit was involved in the design of the study. The meeting with the PPI Panel lasted just over one hour and was conducted by teleconference.

Results: Although we only had a short period of time to explore the study with the PPI Panel, we were able to gain valuable insight into how the trial would be perceived by potential trial participants. Substantive changes were made to the trial to improve the acceptability of the research without compromising the study timelines. Having access to public contributors with relevant lived experience is an important resource for a Clinical Trials Unit and is critical for rapid PPI. The move to remote working due to lockdown required virtual discussions which helped to overcome some of the barriers to organising face-to-face meetings at short notice.

Conclusions: PPI for clinical trials can be conducted in a time-efficient manner within the pressured environment of a pandemic. Involving PPI contributors at an early stage in protocol development maximised the opportunity to shape and influence the trial as well as limited potential delays which could occur if changes to the protocol had to be made at a later stage

Plain English Summary

Patient and public involvement(PPI) describes the active involvement of patients and the public in the research process. Through PPI, patients and members of the public are increasingly involved in the design and conduct of clinical trials. PPI has been shown to improve the quality and relevance of research.

During the COVID-19 pandemic, clinical trials have been playing a vital role in helping us find ways to prevent and treat the infection, and improve our understanding of the virus. It is important that patients and the public are actively involved in deciding how COVID-19 research is carried out. Unfortunately, Research Ethics Committees in the UK have seen far less PPI for COVID-19 research studies compared with research before the pandemic. A key reason for this is that research is being designed much faster than normal, and researchers may feel they do not have time to properly involve patients and the public.

In this paper, we share our experiences of PPI for a COVID-19 clinical trial. We show that it is possible to rapidly involve patients and the public in COVID-19 clinical trials. We also explain how the design of the clinical trial was changed in response to feedback from public contributors. Lastly, we discuss the wider learning from this process which might be useful for researchers planning PPI activities for COVID-19 clinical trials in the future.

Introduction

Patient and public involvement (PPI) in research describes research which is "being carried out 'with' or 'by' members of the public" not just "'to', 'about' or 'for' them"(1). There are compelling moral and practical arguments for promoting active partnerships between researchers and members of the public. The moral argument is that patients and the public have a right to be involved in research that affects them. From a practical perspective, PPI is understood to afford insights to researchers that would not otherwise be available, in order to improve the quality and relevance of studies.

PPI in clinical trials encompasses a range of activities and types of involvement which can occur across all stages of a study from design to dissemination. There is increasing recognition of the value of PPI in clinical trials. A recent systematic review found that PPI is likely to improve participant recruitment levels for clinical trials(2).

Designing and conducting clinical trials in the context of the COVID-19 pandemic presents many challenges which includes ensuring the active involvement of patients and the public in the research process. It is important that these challenges are addressed as research is playing a key role in the global public health response to the pandemic. Well-conducted, high quality clinical trials are critical to both our understanding of the virus and in the development of diagnostic, prophylactic and therapeutic products.

The crisis has sparked a wave of research activity. At the time of submission of this paper, there have been 1728 COVID-19 studies registered on ClinicalTrials.gov since the start of the pandemic (3). In the UK, this has been made possible through the development of new procedures to provide swift reviews by Research Ethics Committees for COVID-19 research. As a result, studies can be reviewed within 24 hours of submission (4).

It is important that with expedited trial design and approvals, PPI in clinical trials is not neglected. Only 20% of the COVID-19 research proposals submitting to Research Ethics Committees (REC) in the UK between 5 March 2020 and 6 April 2020 included PPI compared with 80% of research reviewed prior to the start of the pandemic(5).

The experience of conducting research during the 2014-15 West African Ebola epidemic showed that failure to engage appropriately with communities led to distrust and community backlash which, in turn, hindered early research efforts(6). On the other hand, an understanding of the concerns of communities and an emphasis on building relationships with them based on trust, mutual respect, and active involvement eventually helped to deliver successful Ebola trials (6).

The World Health Organization's Good Participatory Practice Guidelines for Trials of Emerging (and Reemerging) Pathogens emphasises the importance of investing time to effectively engage stakeholders in the research process(7). However, the COVID-19 pandemic represents a time-critical situation. By July 2020, the UK had experienced one of the highest death tolls globally from COVID-19(8). The urgency of the situation demands that research seeking to answer key scientific questions in understanding and tackling the pandemic, including preparing for future outbreaks, be conducted swiftly and efficiently.

In this paper, we present an example of PPI prior to a COVID-19 clinical trial in the UK.

The clinical trial

Patients with cardiovascular disease, high blood pressure and diabetes are at particularly high risk of poor outcomes from COVID-19(9). One possible explanation for this increased risk lies in the medications used in the management of these health conditions. ACE-inhibitors(ACEi) and angiotensin receptor blockers (ARBs) are medications commonly used to treat high blood pressure, diabetes and heart failure. These medications increase the expression of the receptor used by the COVID-19 virus to enter cells in the body. However, the mechanism of this interaction in influencing the severity of illness with COVID-19 is not understood.

RAPID-19 is a pilot open-label randomised trial to establish whether ACEi and ARBs affect the prognosis and severity of illness in ambulatory patients with COVID-19. Patients on ACEi/ARBs who present at hospital with suspected or confirmed COVID-19, and are considered well enough to return home to recover from their COVID-19 symptoms, are eligible to take part in the study. Participants with a confirmed diagnosis of COVID-19 are randomised to either continue taking their medications (control group) or to temporarily withdraw ACEi or ARB for 14 days. Temporary withdrawal of these medications is not associated with clinically significant increases in blood pressure and participants will be monitored for safety concerns. Participants are asked to complete a daily diary for 14 days to monitor their symptoms. They are also followed up by telephone at 7 days, 14 days, and 30 days post-randomisation. Researchers are available via telephone and in person to address concerns and record adverse events.

RAPID-19 is a collaboration between King's Health Partners and the London School of Hygiene & Tropical Medicine (LSHTM) Clinical Trials Unit (CTU). The trial is sponsored by King's College London. In accordance with the INVOLVE statement on public involvement in research and research ethics committee review(10), ethics approval was not sought for our public involvement work on the clinical trial.

Methods

The RAPID-19 Trial Management Group (TMG) began to develop a protocol for the trial which they planned to submit for fast-track ethics review. The LSHTM CTU has an established cardiovascular disease(CVD) PPI panel (the Panel) comprising 10 members who have either lived experience of CVD, or have direct experience as a carer or family member to someone with CVD. The purpose of the Panel is to

work with the CTU in the design of cardiovascular clinical trials to optimise the acceptability, impact and relevance of studies

We had only two working days to explore the RAPID-19 Trial with the Panel and elicit feedback. It was agreed to hold a meeting by teleconference to meet the needs of all attendees. Prior to the meeting, the members were emailed a written summary of the proposed trial design. The meeting was attended by seven members of the Panel and lasted just over one hour. A topic guide was used to facilitate the discussions. The meeting was not audio-recorded. In a separate meeting, we also sought feedback on the trial from a patient representative on the National Cardiac Audit Programme.

Below we describe the insights offered by PPI consultation and the actions taken in response.

The Guidance for Reporting Involvement of Patients and the Public (GRIPP2) short form checklist was used to guide the reporting of PPI in the paper(11).

Results

Risk assessment

A key issue raised by the Panel related to the intervention itself. The intervention violates a common maxim of medical advice, that one should not abruptly stop taking prescribed medications without consulting a doctor. The contributors described their personal experiences of changing or stopping medications. They explained that this decision requires a discussion between their primary care doctor and hospital consultant to examine the impact and possible side effects of the regime change. A decision to stop a medication may require the dose to be reduced slowly over time.

In the trial scenario, the research staff may not have access to the complete medical history of patients they approach, and it may not always be practical to discuss their case with relevant specialists. The Panel explained that in order to consider participating in the trial, they would need to feel confident that the attending clinician who offers recruitment into RAPID-19 has assessed their clinical history and that, taking into account all relevant information, the clinician is able to make an informed assessment as to their suitability for the trial.

Clinical monitoring

PPI contributors remarked how the pandemic is creating an exceptional backdrop for the conduct of research. People are isolated from friends and family, some are fearful of attending hospitals and terrified of how the virus may affect them. Participants will be feeling risk-averse, and consequently need to feel assured that steps have been taken to mitigate any risks to which they could be exposed.

One concern raised by the Panel was that the level of clinical monitoring provided by the trial does not satisfactorily address the risk felt by participants told to stop taking their medication. For example, how would participants know if their blood pressure becomes dangerously high. There were mixed feelings

about the usefulness of home blood pressure monitors being provided to participants. Some felt that this would be an additional burden and may cause participants to feel worried they are not using the device correctly.

Adherence to treatment arm

Members of the Panel commented that some participants may find it difficult adjusting to the change in medication dictated by the intervention. One concern was that participants who receive their medications in pre-prepared dosette boxes from a pharmacy for multidrug regimens would need to identify and remove the correct tablet from their blister pack. It was thought that some patients accustomed to a set routine may forget that they are under a new, altered drug regime and forget to remove the tablet

The Panel also discussed whether automated mobile phone messaging could be used to help participants adhere to their allocated treatment. It was felt that in order to be effective, text messages would need to be sent to coincide with the times participants normally take their tablets and would therefore require customised messaging. The question was raised as to what to do for participants who do not have mobile phones.

Response by TMG to evidence from the PPI Panel

The TMG considered the evidence from the PPI panel and used the insights gained to adapt the trial protocol as follows:

Risk assessment

The TMG added a further exclusion criterion to the eligibility criteria. It supports the case for the attending clinician to assess, in conjunction with the patient and on the basis of the patient's clinical history, whether their participation in the trial would put them at risk. Patients who are considered at significant clinical risk from participation in the trial would not be eligible to take part.

Clinical monitoring

The TMG developed a daily patient diary for participants to use to monitor their symptoms. The records have been designed to alert the participants if they experience any potentially harmful symptoms that develop as a consequence of stopping their medication. It is intended that these alerts are discussed with the clinical team during their daily phone calls.

Adherence to treatment arm

The participant diary was also specifically designed to help participants remember a change in their medication by getting them to complete their diary at the same time each day. The diary asks participants if they have adhered to their treatment allocation. Furthermore, the TMG decided to exclude from the trial those patients who receive medication in a blister pack/dosette box.

Despite the short time-frame available, we were able to gain valuable insight into how the trial would be perceived by potential trial participants. The ability to feed the PPI panel views directly into the protocol, with little delay to the REC application process, indicates that a responsive process is possible and appropriate.

Discussion

Time constraint has been cited as a barrier to PPI in research during the pandemic (12). The experience described in this paper demonstrates that valuable PPI for clinical trials can be conducted in a time-efficient manner within the pressured environment of a pandemic. The adaptations to the trial protocol which were made as a consequence of the PPI work were vital for improving the acceptability of the study to potential participants and which may in turn help to support recruitment and retention into the trial. Key to this process was involving PPI contributors at a very early stage which maximised the opportunity to shape and influence the trial. This also limited potential delays which could occur if changes to the protocol had to be made at a later stage. The approach to PPI employed in this paper can be applied to COVID-19 research conducted in other countries as well as for future epidemics.

The time required to identify appropriate public contributors to involve in the research process at short notice can also be a barrier to PPI. In recognition of this issue, the Health Research Authority has launched a public involvement service for COVID-19 research studies applying for fast-track ethics review(13). The service provides guidance to researchers on who they should involve, what type of involvement they should seek, and then matches research teams with coordinators of relevant public involvement groups across the country.

We were fortunate to already have access to an active PPI panel as well as individual public contributors. The members of our Panel have varying levels of experience with PPI in clinical trials. Some PPI roles are better suited to contributors with previous experience in trials. However, providing feedback on early stage trial development relies heavily on experiential knowledge and we found that a lack of PPI trials experience was not a barrier to contributing to discussions.

In addition, this contribution is enhanced through the development of good working relationships both between the facilitators and PPI members, and between the members themselves which provides the optimal environment for rapid consultation work.

Building strong relationships with public contributors is particularly important during the crisis as it is an anxious and stressful time for all and especially for those groups with underlying health conditions. The majority of members of our Panel have significant health issues and, against the background of the pandemic, the PPI process may be particularly anxiety-provoking for some contributors. Conducting PPI during this time offers opportunities to understand that anxiety, but also requires a mindful, respectful, and emotionally sensitive approach in order to avoid any exacerbation of concerns.

The Evidence Base for Public Involvement in Clinical Trials (EPIC) study explores the different models of PPI that can be applied to a trial(14). The report concludes that responsive and managerial roles for PPI contributors are likely to bring greater benefit to the trial than PPI contributors in oversight positions such as Trial Steering Committees. An experienced PPI contributor appointed to the TMG could also be particularly helpful in advising the TMG on using the feedback from the Panel in adapting the trial protocol.

The Evidence Base for Public Involvement in Clinical Trials (EPIC) study explores the different models of PPI that can be applied to a trial(14). The report concludes that responsive and managerial roles for PPI contributors such as membership of Trial Management Groups(TMGs) are likely to bring greater benefit to the trial than PPI contributors in oversight positions such as Trial Steering Committees. At the time of working on RAPID-19, the CTU did not appoint PPI representatives to TMGs.

On reflection, we recognise that the study would have benefitted from having a PPI representative on the TMG to ensure continual PPI input throughout the development of the trial. Furthermore, the TMG PPI representative could advise the TMG on using the feedback from the Panel in adapting the trial protocol. As a result, the CTU is now looking to expand the models of PPI that we employ and are planning to appoint PPI representatives to sit on the TMG for future trials.

The feedback from the Panel could also be supplemented with a rapid review of existing studies and trials that explore relevant patient acceptability themes applicable to the trial. Additionally, there is now a wealth of on-the-ground experience emerging from COVID-19 clinical trials. New trials would benefit from incumbent research teams sharing examples of good practice in PPI, and how they have overcome challenges to build knowledge and contribute to the evidence-base in this field.

Researchers have been sharing their experiences and best practice for online and remote co-production and patient involvement as well as the pros and cons of the different virtual platforms available(15)(16). We held our meeting by teleconference which unlike with videoconferencing, does not give you the benefit of non-verbal cues and it can be challenging to ensure that everyone in the group has an opportunity to speak. We followed up the meeting with an email to all attendees asking them to phone the facilitator or to send any further comments by email to ensure that we were able to benefit from the contribution of all attendees.

While there are inherent challenges with working virtually, it does offer some opportunities. Even at the best of times, there are a number of difficulties with organising face-to-face meetings at short notice. This includes the availability of some members especially those with caring responsibilities, the cost of travel and the fact that our Panel is geographically dispersed across England. Virtual meetings can offer important solutions to these practical difficulties. In addition, new ways of working may help to broaden the representation of patients and public involved in research and address the lack of diversity of PPI contributors which has long been recognised as a weakness of the PPI system(17).

PPI in research plays an important role in building public trust in scientific evidence(18). A survey conducted by the Academy of Medical Sciences found that only a third of the public trust evidence generated through medical research(18). Crises create a climate of fear and panic, where misinformation and distrust can easily proliferate. Recent scandals during the pandemic, including the well-publicised retraction of the Lancet journal article on hydroxychloroquine(19) may further erode trust in science. Lack of trust may not only compromise recruitment into critical clinical trials, but can also undermine the translation of that evidence into clinical practice. Funders, Sponsors and Research Ethics Committees must uphold PPI standards for COVID-19 research including those for studies eligible for fast-track review.

Limitations

This paper describes the lessons learnt for conducting rapid PPI consultation from the perspective of the research team. However, the PPI Panel have contributed to the manuscript and are named as an author on the paper. The findings presented in this paper could be strengthened by using qualitative research to actively explore with the PPI contributors, their experience of rapid PPI, the perceived impact of their involvement and to consider how they feel the process could be improved.

Conclusion

The need for PPI is arguably more important now than ever. Conducting research in partnership with patients and members of the public will be critical for the successful delivery of COVID-19 clinical trials. The crisis has exposed existing health inequalities associated with socioeconomic status and ethnicity. PPI can help to understand the challenges faced by marginalised groups, and to encourage participation in clinical trials by groups who are typically under-represented, such as ethnic minority communities. This is an opportune time to capitalise on the public's heightened interest in research and science, and to build research literacy which leaves a lasting legacy to the wider society.

Declarations

Acknowledgements

We would like to thank members of the London School of Hygiene & Tropical Medicine Clinical Trials Unit' Research Advisory Group(PPI panel).

Competing interests

The authors declare that they have no competing interests.

Ethics approval and consent to participate

Not applicable

Funding

RAPID-19 is funded by King's Together.

References

- 1. What is public involvement in research? 2015. http:// www.invo.org.uk/find-out-more/what-is-public-involvement-inresearch-2/. Accessed 15 June 2020.
- Crocker JC, Ricci-Cabello I, Parker A, Hirst JA, Chant A, Petit-Zeman S et al. Impact of patient and public involvement on enrolment and retention in clinical trials: systematic review and meta-analysis. BMJ 2018; 363. doi: https://doi.org/10.1136/bmj.k4738
- 3. ClinicalTrials.gov. Accessed 03 August 2020
- Health Research Authority. Fast track review guidance for COVID-19 studies. 2020 https://www.hra.nhs.uk/covid-19-research/fast-track-review-guidance-covid-19-studies/. Accessed 18 June 2020.
- 5. Health Research Authority. Involving the public in COVID-19 research: a guest blog by Bec Hanley and Maryrose Tarpey. 2020. https://www.hra.nhs.uk/about-us/news-updates/involving-public-covid-19-research-guest-blog-bec-hanley-and-maryrose-tarpey/. Accessed 29 June 2020.
- National Academies of Sciences, Engineering, and Medicine. Integrating clinical research into epidemic response: The Ebola experience. Washington, DC: The National Academies Press.2017. doi: https://doi.org/10.17226/24739.
- 7. World Health Organization. WHO, Good participatory practice guidelines for trials of emerging (and re-emerging) pathogens that are likely to cause severe outbreaks in the near future and for which few or no medical countermeasures exist (GPP-EP). 2016.https://www.who.int/blueprint/what/norms-standards/GPP-EPP-December2016.pdf?ua=1 . Accessed 19 June 2020.
- John Hopkins University and Medicine. COVID-19 Dashboard by the Center for Systems Science and Engineering (CSSE) at Johns Hopkins University (JHU). https://coronavirus.jhu.edu/map.html. Accessed 20 July 2020
- Preliminary estimates of the prevalence of selected underlying health conditions among patients with coronavirus disease 2019—United States, Centers for Disease Control and Prevention. February 12–March 28, 2020. MMWR Morb Mortal Wkly Rep. 2020; 69: 382-386. https://www.cdc.gov/mmwr/volumes/69/wr/mm6913e2.htm . Accessed 28 June 2020.
- 10. Health Research Authority / INVOLVE (2016), Public involvement in research and research ethics committee review. invo.org.uk/posttypepublication/patient-andpublic-involvement-in-research-and-research-ethics-committee-review. Accessed 28 June 2020
- 11. Staniszewska S, Brett J, Simera I, et al. GRIPP2 reporting checklists: tools to improve reporting of patient and public involvement in research. BMJ. 2017;358:j3453
- 12. Health Research Authority. HRA COVID-19 Public Involvement workshop. Available on request.

- Health Research Authority New public involvement service for COVID-19 studies. 2020. https://www.hra.nhs.uk/about-us/news-updates/new-public-involvement-service-covid-19-studies/. Accessed 28 June 2020.
- 14. Gamble C, Dudley L, Allam A, Bell P, Buck D, Goodare H, et al. An evidence base to optimise methods for involving patient and public contributors in clinical trials: a mixed-methods study.Health Serv Deliv Res 2015;3(39)
- Digital and remote codesign discussion.
 https://docs.google.com/document/d/1nWKI1G_erwNyI4OuTVQT4vHznx4170PEVD01ioi7xtA/edit .
 Accessed 25 June 2020.
- 16. Farr M, Davies R, Davies P, Bagnall D, Brangan E, Andrews H. (2020). A map of resources for coproducing research in health and social care. National Institute for Health Research (NIHR) ARC West and People in Health West of England; University of Bristol and University of West of England. Version 1.2, May 2020. https://arc-w.nihr.ac.uk/Wordpress/wp-content/uploads/2020/05/Map-ofresources-Web-version-v1.2.pdf . Accessed 25 June 2020.
- 17. Ocloo J, Matthews R. From tokenism to empowerment: progressing patient and public involvement in healthcare improvement. 25(8):626-32. http://dx.doi.org/10.1136/bmjqs-2015-004839
- Academy of Medical Sciences. Enhancing the use of scientific evidence to judge the potential benefits and harms of medicines.2017. https://acmedsci.ac.uk/file-download/44970096. Accessed 28 June 2020.
- 19. Boseley S and Davey M. Covid-19: Lancet retracts paper that halted hydroxychloroquine trials. The Guardian, 4 June 2020. https://www.theguardian.com/world/2020/jun/04/covid-19-lancet-retracts-paper-that-halted-hydroxychloroquine-trials. Accessed 20 June 2020.